Author's response to reviews

Title: Attitudes toward Fertility and Childbearing Scale (AFCS): an assessment of a new Instrument for women not yet mothers in Sweden

Authors:

Malin Soederberg (malin.soderberg@ki.se)
Ingela Lundgren (ingela.lundgren@gu.se)
Kyllike Christensson (kyllike.christensson@ki.se)
Ingegerd Hildingsson (ingeberd.hildingsson@miun.se)

Version: 6 Date: 27 September 2013

Author's response to reviews: see over
Concerning the manuscript,
Attitudes toward Fertility and Childbearing Scale (AFCS): an assessment of a new instrument for women not yet mothers in Sweden

Dear BMC Pregnancy and Childbirth Editorial Team,

I am pleased to inform you that Edanz now has revised the manuscript. A full editorial proof reading has been made. Particular attention has been paid to the abstract, the text format and spelling. I find the result satisfying, and I hope you will as well.

Yours sincerely,

Malin Söderberg
RNRM, PhD
Department of Women’s and Children's Health
Karolinska Institutet
Retziusväg 13 A-B, level 4
SE-171 77 Stockholm
Sweden

+46 8-524 824 16 tel

Corresponding author:
Malin Söderberg
Karolinska institutet
Department of Women's and Children's Health
Retziusväg 13 A-B
S-171 77 Stockholm, Sweden
Phone: +46852482416
Email: malin.soderberg@ki.se